Author's response to reviews

Title: Antiinflammatory evaluation of leaves of Plumeria acuminata

Authors:

M Gupta (malayagupta@yahoo.co.in)
Uk Mazumder (mazumderu@yahoo.co.in)
P Gomathi (pgoms@yahoo.com)
V Thamil Selvan (pharmtamil@yahoo.co.in)

Version: 6 Date: 11 October 2006

Author's response to reviews: see over
Date: 10.10. 06

From,

PROF. Mrs. MALAYA GUPTA,

Division of Pharmacology,

Department of Pharmaceutical Technology,

Jadavpur University,

Kolkata.

India - 700 032.

To,

Editor-in-Chief,

BMC Complementary and Alternative medicine.

Sub: Submission of revised manuscript entitled "Antiinflammatory evaluation of leaves of Plumeria acuminata".

Dear Sir,

Here by I am sending the revised manuscript entitled "Antiinflammatory evaluation of leaves of Plumeria acuminata" for publication in your esteemed Journal. I would be highly obliged if you acknowledge the receipt.

Thanking you,

Yours Faithfully

Malaya Gupta
Answers for the questions asked for authors

Question: In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.

Answer: No

Question: Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.

Answer: No

Question: Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.

Answer: No

Question: Do you have any other financial competing interests? If so, please specify.

Answer: No

Question: Non-financial competing interests
Are there any non-financial competing interests (political, personal, religious, academic, ideological, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.

Answer: No
Replies:

1. As per the suggestion the cover letter was removed from the manuscript.
2. As per the suggestion competing interests was included in the manuscript.
3. As per the suggestion the answers were given for the questions asked for the authors.